Author’s response to reviews

Title: Descending Necrotizing Mediastinitis Caused by Streptococcus Constellatus in an Immunocompetent Patient: Case Report and Review of the Literature

Authors:

rui-hai ye (yehoo2003@163.com)

jun chao yang (yangjunchaozj@163.com)

hui hua hong (honghuiha@sina.com)

yu ping mao (2242723113@QQ.com)

yuan hong zhu (zhuyuan0487@126.com)

yu cao (caoyumail@163.com)

zhen wang (wangzhen610@sina.cn)

Version: 1 Date: 10 Jun 2019

Author’s response to reviews:

Dear editors,

We would like to submit the manuscript with the following title and authors for publication in BMC PULMONARY MEDICINE.

Title: Descending Necrotizing Mediastinitis Caused by Streptococcus Constellatus in an Immunocompetent Patient: Case Report and Review of the Literature

Authors: Rui-hai Ye, Jun-chao Yang, Hui-hua Hong, Yu-ping Mao, Yuan-hong Zhu, Yu Cao and Zhen Wang

BMC pulmonary medicine provide abundant clinical cases for clinical medical workers. It is a excellent medical journal especially about case report. We are deeply impressed by your earnest, rigorous and responsible scientific attitude. We have supplemented relevant ethical information as required. The health of patients is our mission. The seriousness of your magazine is worthy of our study. A small step in our medical clinical case that is a big step in the history of clinical medicine. We declare that submitted manuscript does not contain previously published material, and is not under consideration for publication elsewhere. The manuscript is truthful original work without fabrication, fraud or plagiarism. Each author has made an important scientific contribution to the study and is thoroughly familiar with the primary data. All authors listed have read the complete manuscript and have approved submission of the paper. All authors declare that there is no conflict interest.

We would be happy if the manuscript will be evaluated by your Editorial Board members for publication in BMC PULMONARY MEDICINE.
With best regards,

Yours sincerely,

Zhen Wang
Department of Respiratory Medicine, Zhejiang Province Hospital of Traditional Chinese Medical, China,